Author’s response to reviews

Title: Development of anthracycline-induced dilated cardiomyopathy due to mutation on LMNA gene in a breast cancer patient: a case report

Authors:

Jock Chichaco Kuruc (jockrandolph@gmail.com)
Armando Durant Archibold (armando.durant@gmail.com; adurant@indicasat.org.pa)
Jorge Motta (drjmotta@gmail.com)
K.S. Rao (jrao@indicasat.org.pa)
Barry Trachtenberg (brachtenberg@houstonmethodist.org)
Carlos Ramos (laito52@yahoo.com)
Hongyu Wang (Hongyu.Wang@uth.tmc.edu)
David Gorenstein (david.gorenstein@am-biotech.com)
Fredrik Vannberg (fredrik.vannberg@biology.gatech.edu)
King Jordan (king.jordan@biology.gatech.edu)

Version: 2 Date: 30 Jun 2019

Author’s response to reviews:

June 30, 2019

Dr. Spenceley Jones
BMC Cardiovascular Disorders

Dear Dr Spenceley Jones,

I hereby submit a revised version of the manuscript BCAR-D-19-00138R1, Development of anthracycline-induced dilated cardiomyopathy due to mutation on LMNA gene in a breast cancer patient: a case report. Thank you for giving us the opportunity to revise and resubmit this manuscript. I appreciate the time and valuable comments provided by the Editor, and we have incorporated the suggested changes into the manuscript, and I am sending the information requested by the Editor.
I have responded to the suggestions below. To make the changes easier to identify where necessary, I have numbered them.

Editor:

1. We note that the order of authors has changed since the original submission of the manuscript. Please fill out a change in authorship form and send it to the Editorial office to be approved by the Editor. The form can be found here: https://www.biomedcentral.com/getpublished/editorial-policies#authorship.

R. We are sending the authorship form requested. Unfortunately, I made a mistake at the moment of writing the information related with the order of the authors. Because of this I did the correction. Since I represent all the authors of this manuscript as corresponding author, based on the agreement among all authors, I am signing the Authorship Change Form on behalf of all authors. The form, with all the information requested, has been sent by email to the Editorial Office.

2. Please could you clarify if a copy of the consent to publish form is available to the Editor.  
R. The consent form for publishing is available to the Editor.

3. Please ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.  
R. Pursuant to the suggestion made by the Editor we cited Figure 1 within the text (page 4, line 90).

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary files are cited within the text.  
R. I am submitting the manuscript following the instructions given by the Editor.

5. Reviewer reports: If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English.  
R. There was not a request of English revision made by the reviewers of this manuscript.

Thank you again for consideration of our revised manuscript.

Best regards,

Armando Durant Archibold, PhD
Biochemistry and Molecular Medicine Research Unit
Center for Drug Discovery and Biodiversity
Clayton, City of Knowledge |Building 208 |Panama City, Panama
Phone: +507 5170715  Fax: +507 5170000  | adurant@indicasat.org.pa
Website: www.indicasat.org.pa